Author’s response to reviews

Title: Prevalence of Trichomonus Vaginalis Infection in Egyptian Women: Cross-Sectional Study

Authors:

Ahmed Mahmoud MD (ahmad.mahmoud.ali@hotmail.com)
Nadine A Sherif MD (nadine.sherif@kasralainy.edu.eg)
Rana Abdella MD (rana.abdella@yahoo.com)

Version: 2 Date: 25 April 2014

Author’s response to reviews:

• Page numbering done
• Line numbering done
• Abstract formatting: into background- Results- Conclusion
• Conclusion title included separately
• Competing interests added
• Authors’ contribution added
• Acknowledgement added
• Funding added